Author's response to reviews

Title: Health care seeking among pulmonary tuberculosis suspects and patients in rural Ethiopia: a community-based study

Authors:

Solomon Yimer (yimsolo@yahoo.com)
Carol Holm-Hansen (carol.holm-hansen@fhi.no)
Tilahun Yimaldu (tilahunyimaldu2003@yahoo.it)
Gunnar Bjune (g.a.bjune@medisin.uio.no)

Version: 5 Date: 20 November 2009

Author's response to reviews:

Dear Production Editor,

We would very much like to thank the Editorial Production Team for their comments. As per your comments, we have made the required formatting changes and final revisions to our manuscript. The changes include:

1. The figures referenced in the manuscript text as “Figure 1a” and “Figure 1b” are now referenced as Figure 1 and Figure 2 throughout the manuscript.
2. The city is already included in the affiliation detail
3. We have also made some corrections/changes in the text.

With Best Regards,
Solomon Abebe Yimer
Corresponding author